

REVIEW HISTORY

Paper: **“Appropriate Processing Time: Valuing Process over Product”**

Corresponding Author: Beth Garcia

Email: egarcia@wtamu.edu

Doi: 10.19044/ejes.v6no3a1

Peer review:

Reviewer 1: Mary Ann Hollingsworth

University of West Alabama, Livingston, Alabama, USA

Reviewer 2: Blinded

Published: 31.09.2019